{\bf name}       & string               & \vvar{loiter-west-zone}  & yes   & -           & \pageref{sec_bhv_name}      \\ 
duration         & double               & \vvar{600}               & -     & \var{-1}    & \pageref{sec_duration} \\ 
duration\_status & \vvar{MOOSVAR}       & \vvar{loiter\_remaining} & yes   & -           & \pageref{sec_duration} \\ 
priority, pwt    & double               & \vvar{100}          & -          & \vvar{100}  & \pageref{sec_bhv_priority}       \\ 
runflag          & \vvar{MOOSVAR=value} & \vvar{LOITERING = maybe} &  yes  & -           & \pageref{sec_flags} \\ 
endflag          & \vvar{MOOSVAR=value} & \vvar{LOITERING = done} &  yes   & -           & \pageref{sec_flags} \\ 
activeflag       & \vvar{MOOSVAR=value} & \vvar{LOITERING = yes} &  yes    & -           & \pageref{sec_flags} \\ 
inactiveflag     & \vvar{MOOSVAR=value} & \vvar{LOITERING = off} &  yes    & -           & \pageref{sec_flags} \\ 
idleflag         & \vvar{MOOSVAR=value} & \vvar{LOITERING = no} &  yes     & -           & \pageref{sec_flags} \\ 
nostarve         & \vvar{MOOSVAR},double & \vvar{INFO,60}     &  yes       & -           & \pageref{sec_stalevars} \\ 
perpetual        & Boolean string        & \vvar{false}        &  no       & \vvar{false} & \pageref{sec_perpetual} \\ 
post\_mapping    & \vvar{MOOSVAR,MOOSVAR} & \vvar{FOO,BAR}    & yes        & -            & \pageref{sec_bhv_post_mapping} \\
updates          & \vvar{MOOSVAR}       & \vvar{LOITER\_INFO} &  yes       & -            & \pageref{sec_updates}  \\ 
condition        & Logic Expression     & \vvar{QUALITY <= 7} &  yes & -   & \pageref{sec_conditions} \\ \hline \hline
